

British Society for Rheumatology RHEUMATOLOGY **ADVANCES IN PRACTICE** 

# Guideline

# Pain management in people with inflammatory arthritis: **British Society for Rheumatology guideline scope**

Ian C. Scott (1,2,\*, Opeyemi Babatunde (1, Christopher Barker<sup>3</sup>, Rebecca Beesley<sup>4</sup>, Richard Beesley<sup>4</sup>, Hollie Birkinshaw<sup>5</sup>, Mel Brooke<sup>6</sup>, Hema Chaplin <sup>7,8</sup>, Lara Chapman <sup>9</sup>, Coziana Ciurtin (1)<sup>10</sup>, James Dale<sup>11</sup>, Dervil Dockrell (1)<sup>12</sup>, Emma Dures<sup>13</sup>, Kathyrn Harrison<sup>14</sup>, Meghna Jani 10<sup>15,16</sup>, Charlotte Lee<sup>17</sup>, Maura McCarron<sup>18,19</sup>, Christian D. Mallen 10<sup>1</sup>, Assie O'Connor<sup>2</sup>, Claire Pidgeon<sup>20</sup>, Tamar Pincus<sup>17</sup>, Dee Pratt<sup>21</sup>, Yeliz Prior <sup>22</sup>, Karim Raza<sup>23,24</sup>, Zoe Rutter-Locher 10<sup>25</sup>, Seema Sharma 10<sup>26</sup>, Katie Shaw<sup>27</sup>, Samantha Small<sup>28</sup>, Tilli Smith<sup>1</sup>, Lesley Tiffin<sup>29</sup>, Jordan Tsigarides (1)<sup>30,31</sup>, Mikalena Xenophontos<sup>32,33</sup>, Nicholas G Shenker (1)<sup>34,35</sup> <sup>1</sup>Primary Care Centre Versus Arthritis, School of Medicine, Keele University, Staffordshire, UK <sup>2</sup>Haywood Academic Rheumatology Centre, Midlands Partnership University NHS Foundation Trust, Staffordshire, UK <sup>3</sup>Community Pain Management Service, Mersey Care NHS Foundation Trust, Mersey, UK <sup>4</sup>Juvenile Arthritis Research (JAR), Tonbridge, UK <sup>5</sup>School of Primary Care. Population Sciences and Medical Education, Primary Care Research Centre, University of Southampton, Southampton, UK <sup>6</sup>Expert by experience, Bath, UK <sup>7</sup>Health Psychology Section, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK <sup>8</sup>School of Health and Society, Centre for Applied Health Research, University of Salford, Salford, UK <sup>9</sup>Leeds Institute of Rheumatic and Musculoskeletal Medicine, University of Leeds, Leeds, UK <sup>10</sup>Division of Medicine, Centre for Adolescent Rheumatology, University College London, London, UK <sup>11</sup>Department of Rheumatology, NHS Lanarkshire, Lanarkshire, UK <sup>12</sup>Bone Research Group, University of Edinburgh and NHS Lothian, Edinburgh, UK <sup>13</sup>School of Health and Social Wellbeing, University of the West of England (UWE), Bristol, UK <sup>14</sup>Department of Paediatric Rheumatology, Birmingham Children's Hospital, Birmingham, UK <sup>15</sup>Centre for Epidemiology Versus Arthritis, Centre for Musculoskeletal Research, The University of Manchester, Manchester, UK <sup>16</sup>NIHR Manchester Biomedical Research Centre, Manchester University NHS Foundation Trust, Manchester Academic Health Science Centre, Manchester, UK <sup>17</sup>School of Psychology, University of Southampton, Southampton, UK <sup>18</sup>Department of Rheumatology, Belfast Health and Social Care Trust, Belfast, UK <sup>19</sup>School of Psychology, Queen's University Belfast, Belfast, UK <sup>20</sup>Department of Paediatric Occupational Therapy, Birmingham Women's and Children's NHS Foundation Trust, Birmingham, UK <sup>21</sup>Department of Musculoskeletal Physiotherapy, Surrey Downs Health and Care Community Services, Epsom and St Helier University Hospitals NHS Trust, Surrey, UK <sup>22</sup>Centre for Human Movement and Rehabilitation, School of Health and Society, University of Salford, Salford, UK <sup>23</sup>Department of Rheumatology, Hywel Dda University Health Board, Wales, UK <sup>24</sup>Rheumatology Research Group, Institute of Inflammation and Ageing, University of Birmingham, Birmingham, UK <sup>25</sup>Department of Rheumatology, Guy's and St Thomas' NHS Trust, London, UK <sup>26</sup>Department of Rheumatology, Manchester University NHS Foundation Trust, Manchester, UK <sup>27</sup>Department of Physiotherapy, Bolton NHS Foundation Trust, Bolton, UK <sup>28</sup>Department of Paediatric Rheumatology, Southampton Children's Hospital, Southampton, UK <sup>29</sup>Department of Rheumatology, Newcastle Hospitals NHS Foundation Trust, Newcastle, UK <sup>30</sup>Department of Rheumatology, Norfolk and Norwich Hospital, Norwich, UK <sup>31</sup>Norwich Medical School, University of East Anglia, Norwich, UK <sup>32</sup>Kennedy Institute of Rheumatology, University of Oxford, Oxford, UK <sup>33</sup>Department of Rheumatology, Oxford University Hospitals, Oxford, UK <sup>34</sup>Department of Rheumatology, Cambridge University Hospitals, Cambridge, UK <sup>35</sup>Department of Medicine, University of Cambridge, Cambridge, UK \*Correspondence to: Dr Ian C. Scott, Primary Care Centre Versus Arthritis, School of Medicine, Keele University, Keele, UK. E-mail: i.scott@keele.ac.uk

Received: 3 July 2024. Accepted: 4 October 2024

© The Author(s) 2024. Published by Oxford University Press on behalf of the British Society for Rheumatology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

#### **Executive Summary**

Pain is a common symptom in people with inflammatory arthritis (IA), which has far-reaching impacts on their lives. Recent electronic health record studies demonstrate that UK-based pain care in people with IA commonly involves the prescribing of long-term opioids and gabapentinoids, despite an absence of trial evidence for their efficacy. Patient surveys suggest that non-pharmacological pain management is underused. A UK-specific guideline on pain management for people with IA is required to resolve this. This scoping document outlines the context and prioritized clinical questions for the first British Society for Rheumatology (BSR) guideline on pain management for people with IA. The guideline aims to provide evidence-based recommendations on how pain can be best managed in people with IA (including its assessment, and pharmacological and non-pharmacological treatments), ensuring that people with IA in the UK are offered evidence-based pain management strategies. The guideline is for healthcare professionals involved in the care of people with IA of all ages and genders, people with IA and their families and carers, NHS managers and healthcare commissioners, and other relevant stakeholders such as patient organizations. It will be developed using the methods outlined in the BSR's 'Creating Clinical Guidelines' protocol.

#### Lay Summary

#### What does this mean for patients?

Inflammatory arthritis includes conditions causing swollen and painful joints. The main types are rheumatoid arthritis, psoriatic arthritis, axial spondyloarthritis and juvenile idiopathic arthritis. It is common, affecting over 1 in 100 adults and 4 in 10 000 children and young people in the UK. Despite the availability of effective medicines to reduce joint inflammation, many people with inflammatory arthritis continue to experience daily pain. This has a great impact on their lives. They often receive strong pain medicines called opioids, despite little evidence that they help pain in inflammatory arthritis. The British Society for Rheumatology is the UK's main medical society for healthcare professionals providing care to people with inflammatory arthritis. It provides guidelines that recommend how healthcare professionals can best treat people with specific conditions. This paper outlines the plan to develop the first guideline from the British Society for Rheumatology about how pain should be managed in people with inflammatory arthritis. It will be developed by a group of people from many backgrounds (including doctors, nurses, physiotherapists, podiatrists, occupational therapists, psychologists, pharmacists and people with inflammatory arthritis) and will be based on the best available research, identified by reviewing published studies in a systematic way.

Keywords: inflammatory arthritis, rheumatoid arthritis, psoriatic arthritis, axial spondyloarthritis, juvenile idiopathic arthritis, pain, analgesic, pharmacological, non-pharmacological

#### Why the guideline is needed

Pain is a common problem for people with inflammatory arthritis (IA), with many experiencing daily pain and reporting dissatisfaction with their arthritis pain [1, 2]. UK-based electronic health record studies indicate that pain care in people with IA commonly involves the prescribing of long-term opioids and gabapentinoids [3, 4], despite an absence of evidence for efficacy [5] and potential adverse events [6, 7]. UK-based patient surveys also suggest that non-drug IA pain care (such as physiotherapy-supported exercise regimens and orthoses) are underused [8, 9]. A guideline from the British Society for Rheumatology (BSR) on pain management in people with IA is needed to improve pain and its care in people with IA. By providing recommendations developed through multi-stakeholder involvement (including healthcare professionals who care for people with IA and people with IA) based on the best available evidence, it will support the provision of evidence-based pain management to people with IA treated in the UK.

#### Key facts and figures

Recent studies using national electronic health record data from primary care show that IA is common in adults, with over 1% of people aged  $\geq$ 18 years, and over 2.5% of those aged >65 years meeting criteria for a diagnosis of rheumatoid arthritis (RA), psoriatic arthritis (PsA) or axial spondyloarthritis (AxSpA) in England in 2020 [10]. Juvenile idiopathic arthritis (JIA) is rarer, with <0.1% of children and young people having a validated primary care recorded diagnosis of JIA in 2018 [11], although the condition can continue into adulthood [12]. Optimizing pain care in people with IA will therefore improve the lives of a substantial number of people across the UK.

Current rheumatology care for adults with IA focuses on delivering 'treat-to-target' strategies, in which disease activity is regularly measured [e.g. using the Disease Activity Score for 28 Joints (DAS28) in those with RA] and disease-modifying anti-rheumatic drugs (DMARDs) escalated until the target of remission or low disease activity is achieved. In children and young people treat-to-target approaches are also used, although they are less well defined and trial evidence comparing this approach with another or no strategy are lacking [13]. Treat-to-target has transformed many outcomes in people with IA, leading to lower disability levels, less radiological damage and improved quality of life [14, 15]. However, while it also improves pain in people with IA [16, 17], it does not fully control it, even in those achieving these targets [18]. The limitations of DMARDs for pain are shown in observational studies, including in the BSR Biologics Registry (in which 79% of people with RA receiving biologic DMARDs belonged to a 'persistent pain' trajectory [19]) and clinical trials [20]. While cohorts of people with early RA show lower disease activity levels and better physical quality of life in the treat-to-target era, pain levels remain similar in pre- and post-2002 cohorts [21].

Longitudinal studies show that many people with IA suffer from persistent pain, including those receiving high-cost biologic DMARDs [19, 22–24]. The impact of pain is far-reaching, being the dominant predictor of psychosocial health [25], independently predicting work disability [26], and being associated with worse quality of life, functioning, mental health, fatigue and well-being [27–30]. Consequently, people with IA consistently rate pain and improved pain care as a key priority [31–33]. Pain management in people with IA is complicated by its multidimensional and multifactorial nature, with nociceptive pain (from actual/threatened damage to non-neural tissue and nociceptor activation e.g. from synovitis), neuropathic pain (from a lesion/disease of the somatosensory nervous system) and nociplastic pain (from altered nociception in the absence of tissue damage/inflammation or a somatosensory nervous system lesion/disease), all playing contributing roles [23, 34, 35]. Nociplastic pain is particularly common, with a systematic literature review and meta-analysis of the prevalence of fibromyalgia in people with RA, PsA and ankylosing spondylitis (a subtype of AxSpA) reporting a pooled prevalence of 21%, 18% and 13%, respectively [35].

#### **Current practice**

Recent UK-based electronic health record studies demonstrate that pain care for people with IA focuses on the prescribing of analgesics [3, 4, 36]. For example, in one study examining the annual prevalence of analgesic prescriptions in the Clinical Practice Research Datalink Aurum (a large database of routinely collected data from >1400 general practitioner (GP) practices spanning 20% of England), the annual prevalences of analgesic, non-steroidal anti-inflammatory drug (NSAID), opioid and gabapentinoid prescriptions in 2020 were 64.5, 22.3, 39.0 and 9.9 per 100 person-years, respectively [3]. This is despite limited trial evidence of efficacy (except for NSAIDs in AxSpA) [5, 37, 38], but many potential harms (including fractures and overdose with opioids and upper gastrointestinal complications with NSAIDs) [6, 39]. Published data on the extent to which non-drug therapies are used for IA pain management in the UK are limited, but suggest they are underused. In a 2009 survey of 1400 people with RA in England 44%, 40%, 28% and 13% reported having received NHS-based physiotherapy, occupational therapy, podiatry and orthoses, respectively [9]. A more recent survey (in 2022) of people with AxSpA reported that, among 294 people diagnosed in the previous 5 years, only 40% stated they had receiving information about physical exercises tailored specifically to their AxSpA in the 12-months postdiagnosis [8]. In the whole sample of 913 people, while 82% rated 'advice on how to manage pain levels' as 'very important', only 39% rated their experience of this as 'very/quite positive'.

#### **Equity considerations**

Health Survey for England 2017 data demonstrate that, in the general population, people are more likely to experience chronic pain (pain that persists or recurs for more than 3 months) if they live in deprived communities, are from minoritized ethnic backgrounds, are female, have more than two long-term conditions or are older [40]. Similarly, in Scottish Health Survey 2022 data, the proportions of people with chronic pain were higher in females, older people and in those living in more deprived areas [41]. There is also evidence that in the general population, opioid use generally [42], and in people with chronic non-cancer pain long-term opioid use specifically [43], is commoner in more deprived areas. These factors are also likely to be important in people with IA, with the annual prevalence of opioid or gabapentinoid prescriptions higher in people with IA living in England that are older, female or living in areas of deprivation and North England [3]. More recently, in 2023, the Arthritis and Musculoskeletal Alliance conducted a national inquiry into musculoskeletal health inequalities, concluding that in patients with inflammatory rheumatic conditions (such as IA), low socioeconomic status is associated with increased pain and poorer outcomes, for which there is no biological basis [44].

### **Previous guidance**

While many contemporary guidelines exist for the management of IA, these primarily focus on reducing disease activity Downloaded from https://academic.oup.com/rheumap/article/8/4/rkae128/7903098 by guest on 06 December 2024

with DMARDs, providing little guidance on pain. For example, in National Institute for Health and Care Excellence (NICE) RA guidelines, the few pain-specific recommendations are that short-term NSAIDs and hand exercises can be considered, periodic multidisciplinary team assessments should occur, people should have access to specialist physiotherapy to learn about transcutaneous electrical nerve stimulators and wax baths for short-term pain relief, and early surgical reviews should be offered for persistent pain from joint damage [45]. This is because these guidelines have focused on pain related to disease activity and are targeted at secondary care services. Similar approaches have been used in NICE guidelines related to PsA and other types of peripheral spondyloarthritis (SpA) [46]. Only one contemporary guideline (from the European Alliance of Associations for Rheumatology [EULAR]) primarily addresses pain management in people with IA [47]. This guideline does not consider JIA. While providing useful information for clinical practice, it is based on a relatively historic umbrella review (considering systematic reviews until 2015), with many systematic reviews published since. It also does not consider pharmacological treatments in detail (with its underpinning umbrella review evaluating non-drug care only) and combines recommendations for both IA and osteoarthritis (whose models of care differ). The 2021 NICE guideline on chronic pain provides general advice on pain management for chronic primary pain (in which no underlying condition adequately accounts for the pain or its impact) [48]. When chronic primary and secondary pain co-exist (which can often occur in people with IA), it recommends using clinical judgement to inform shared decision-making about the management options in the relevant NICE guideline. As NICE guidelines for IA do not focus on pain, healthcare professionals and people with IA may consequently feel uncertain as to the best way to manage chronic pain in this condition.

### What the guideline will cover

#### Who are the target users of the guideline?

The guideline is for all healthcare professionals involved in the management of people with IA, people with IA and their families and carers, NHS managers and healthcare commissioners, and other stakeholders such as patient organizations.

#### Which population will the guideline apply to?

It will cover pain management in adults, children and young people with IA [including RA, PsA and other forms of peripheral SpA, AxSpA, JIA and undifferentiated IA]. All ages and genders will be considered.

#### Settings

The guideline will be of relevance to all UK healthcare settings (primary/community, secondary and tertiary care).

#### Areas that will be covered

The definition of pain from the International Association for the Study of Pain will be used, which defines it as 'an unpleasant sensory and emotional experience associated with, or resembling that associated with, actual or potential tissue damage' [49]. The guideline will consider literature in three key areas relevant to pain in people with IA: (1) the assessment of pain; (2) pharmacological interventions for pain; and (3) non-pharmacological interventions for pain. It will consider evidence for all types of pain (nociceptive, neuro-pathic and nociplastic) [50].

#### Areas that will not be covered

The guideline will not consider the following areas: (1) acute pain due to clearly defined non-rheumatological issues (e.g. infection, post-operative pain, trauma); (2) pain-management procedures (e.g. surgical interventions, vagal nerve stimulation or intra-articular steroids); (3) public health interventions for pain; (4) pain management in people with rheumatological conditions where IA can occur, but is not traditionally considered to fall within the umbrella-term of 'IA' [such as crystal arthritis, connective tissue diseases (including systemic lupus erythematosus), vasculitis and autoinflammatory syndromes]-although the findings from this guideline are likely to be relevant to people with these conditions; and (5) foot pain (being addressed in BSR's 'management of foot health in people with inflammatory arthritis' guideline) [51]. It will also not cover non-pain aspects of IA care, which are addressed in other national guidelines from NICE and BSR.

### Key issues and clinical questions

The following clinical questions have been identified by the Guideline Working Group to address within the guideline. Where relevant, these have been framed in the Population, Intervention, Comparator and Outcome format.

#### Assessment of pain

In people with IA (population):

- Which pain factors (e.g. sensitization) and pain-related factors (e.g. mood) should be considered when assessing pain?
- Which interpersonal consultation-based factors should be considered when assessing pain?
- Which outcome measures should be used when assessing pain?
- Should pain assessments be in-person or remote?
- How often should pain assessments take place?

### Pharmacological treatments for pain

In people with IA (population), which of the following pharmacological treatments (intervention) relative to placebo or other pharmacological/non-pharmacological treatments (comparator), improve pain (primary outcome) and what are their effects on quality of life, function, stiffness, adverse events, analgesic use, sleep, fatigue, mental health, education and employment (secondary outcomes):

- Analgesics—paracetamol, oral NSAIDs, topical NSAIDs, nefopam, opioids.
- Neuromodulators—anti-depressants, gabapentinoids, topical capsaicin, cannabinoids.
- Immunosuppressants—systemic glucocorticoids, conventional synthetic (cs)DMARDs, biologic (b)DMARDs and targeted synthetic (ts)DMARDs.

### Non-pharmacological treatments for pain

In people with IA (population), which of the following nonpharmacological treatments (intervention) relative to placebo or other non-pharmacological/pharmacological treatments (comparator), improve pain (primary outcome) and what are their effects on quality of life, function, stiffness, adverse events, analgesic use, sleep, fatigue, mental health, education and employment (secondary outcomes):

- Exercise and physical activity.
- Psychological therapies.
- Ergonomics.
- Orthotics (excluding orthoses for foot pain, which are considered in the BSR management of foot health in people with IA guideline).
- Other: education; weight management and diet; sleep and fatigue management; digital technologies; complementary therapies; medical devices; and support from others.

# Service organization and delivery within NHS England and devolved nations

People with IA require holistic pain care (reflecting the multidimensional and multifactorial nature of pain) and may present with pain-related symptoms in all healthcare settings (including primary care, where practitioners may feel less confident in assessing disease activity). Collaborative working across healthcare boundaries, and ensuring equitable access to MDT services, is therefore crucial to providing evidence-based, equitable pain management. Recent decades have seen an erosion in the rheumatology MDT, with few UK rheumatology departments having a full complement of healthcare professionals, and the composition and staffing levels of the rheumatology MDT varying substantially by geographical location [52]. Consequently, the Guideline Working Group will provide commissioner-friendly dissemination materials, engage with Specialist Societies and Royal Colleges in guideline implementation and develop relevant audit and quality improvement tools to measure and improve pain care across the healthcare system.

### **Guideline working group**

The multi-disciplinary Guideline Working Group comprises the following members from across the UK: Ian C. Scott (co-chair, Rheumatologist); Nicholas G. Shenker (co-chair, Rheumatologist); Opevemi Babatunde (Evidence Synthesis Researcher); Christopher Barker (General Practitioner); Rebecca Beesley (patient and charity representative); Richard Beesley (carer and charity representative); Hollie Birkinshaw (Research Psychologist); Mel Brooke (patient representative); Hema Chaplin (Chartered Psychologist); Lara Chapman (Podiatrist); Coziana Ciurtin (Adolescent and Adult Rheumatologist and BSR Standards, Audit, and Guidelines Working Group liaison); James Dale (Rheumatologist); Dervil Dockrell (Occupational Therapist); Emma Dures (Chartered Psychologist); Kathyrn Harrison (Paediatric Rheumatologist); Meghna Iani (Rheumatologist); Charlotte Lee (Research Psychologist); Maura McCarron (Rheumatology Specialty Doctor); Christian Mallen Practitioner); Assie O'Connor (Rheumatology (General Pharmacist); Claire Pidgeon (Paediatric Occupational Therapist); Tamar Pincus (Chartered Psychologist); Dee Pratt (Physiotherapist); Yeliz Prior (Occupational Therapist); Karim Raza (Rheumatologist); Zoe Rutter-Locher (Rheumatology Specialist Registrar); Seema Sharma (Rheumatology Specialist Registrar); Katie Shaw (Physiotherapist); Samantha Small (Paediatric Rheumatology Nurse Specialist); Tilli Smith (Evidence Synthesis Researcher); Lesley Tiffin (Nurse Practitioner); Jordan Tsigarides (Rheumatology Specialist Registrar); and Mikalena Xenophontos (Rheumatology Specialist Registrar).

#### Dissemination

The guideline will be accompanied by infographic and video summaries in both patient-friendly and commissionerfriendly formats. The Guideline Working Group will engage with Specialist Societies, Royal Colleges and patient associations to disseminate the guideline to their target audiences. Following publication, the Guideline Working Group will seek to activate local champions within the healthcare community, including BSR's Pain Special Interest Group, to optimize guideline uptake, facilitated by the guideline's audit and quality improvement tools. The guideline is expected to be published in 2026.

#### **Data availability**

No new data were generated or analysed in support of this work.

## Authors' contributions

I.C.S. and N.G.S. drafted this article. All authors contributed to developing the guideline scope, provided critical manuscript editing and revision, and approved the final submitted version.

#### Funding

This work was supported by the British Society for Rheumatology. The guideline's underpinning systematic review is funded by the National Institute for Health and Care Research (NIHR) [Advanced Research Fellowship (NIHR300826 to I.C.S.)]. The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care.

Disclosure statement: C.B. has received: (1) honoraria from Flippin' Pain for public health work; (2) consultancy fees from NICE for providing pain consultancy to NICE Advice in relation to a pharmaceutical company for a clinical trial. C.B. is company director for Pain Recovery Partnership Ltd. This company has not traded since incorporation in 2020. C. C. has received: (1) a grant from GSK (MRC-EMINENT Funding scheme) (paid to employer); (2) honoraria from UCB and Novartis speaker bureau. C.C. is doing research in another disease in collaboration with GSK. J.D. has received speaker fees from Abbvie for presenting at a satellite symposium during BSR conference 2022. C.D.M. is Director of the NIHR School for Primary Care Research. K.R. is an investigator on a clinical trial funded by Bristol Myers Squibb and has received honoraria from Abbvie and Sanofi. I.C.S.'s NHS trust has received research fees to support delivery of commercial trials for various drug therapies. I.C.S. was principal investigator at MPFT for the RA-BE-REAL study (a commercially funded, portfolio-adopted observational study) and is funded by an NIHR Advanced Research Fellowship [NIHR300826]. The views expressed are those of the author (s) and not necessarily those of the NIHR or the Department of Health and Social Care. N.G.S. has received: (1) conference fees from Eli Lilly for attendance, travel and accommodation to two BSR conferences; (2) speaker fees from Eli Lilly for a sponsored lecture. The NIHR Cambridge Biomedical Research Centre (BRC) is a partnership between Cambridge University Hospitals NHS Foundation Trust and the University of Cambridge, funded by the NIHR; NGS is supported by the NIHR Cambridge BRC. The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care. J.T. has received: (1) conference fees from Novartis to attend the ASAS Axial SpA Meeting in Brussels (2022); (2) Honoraria from Novartis for supporting Virtual Reality and Future of Rheumatology sessions at their Hi-5 event. M.B. is a selfemployed patient and public engagement consultant, working with a variety of organizations, charities and p/t contracts. Some of these organizations may have attracted funding from companies manufacturing therapies or treatments for the conditions covered by this guideline, but M.B. has not received direct funding related to this. M.J. is funded by an NIHR Advanced Fellowship [NIHR301413]. The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care. All other authors have declared no conflicts of interest.

#### References

- Strand V, Wright GC, Bergman MJ, Tambiah J, Taylor PC. Patient expectations and perceptions of goal-setting strategies for disease management in rheumatoid arthritis. J Rheumatol 2015; 42:2046–54.
- 2. Taylor P, Manger B, Alvaro-Gracia J *et al.* Patient perceptions concerning pain management in the treatment of rheumatoid arthritis. J Int Med Res 2010;38:1213–24.
- 3. Scott IC, Whittle R, Bailey J *et al.* Analgesic prescribing in patients with inflammatory arthritis in England: observational studies in the Clinical Practice Research Datalink. Rheumatol Oxf Engl 2024;63:1672–81.
- Scott IC, Bailey J, White CR, Mallen CD, Muller S. Analgesic prescribing in patients with inflammatory arthritis in England: an observational study using electronic healthcare record data. Rheumatol Oxf Engl 2022;61:3201–11.
- Whittle SL, Richards BL, Husni E, Buchbinder R. Opioid therapy for treating rheumatoid arthritis pain. Cochrane Database Syst Rev 2011;9:CD003113.
- Chou R, Turner JA, Devine EB *et al*. The effectiveness and risks of long-term opioid therapy for chronic pain: a systematic review for a national Institutes of Health Pathways to Prevention Workshop. Ann Intern Med 2015;162:276–86.
- Muller S, Bailey J, Bajpai R *et al*. Risk of adverse outcomes during gabapentinoid therapy and factors associated with increased risk in UK primary care using the clinical practice research datalink: a cohort study. PAIN 2022;165:2282–90.
- National Axial Spondyloarthritis Society. New research from NASS reveals what patients with axial SpA value during diagnosis, treatment and care. https://nass.co.uk/news/new-research-fromnass-reveals-what-patients-with-axial-spa-value-during-diagnosistreatment-and-care/ (15 September 2024, date last accessed).
- National Audit Office (NAO). Services for people with rheumatoid arthritis—NAO report, 2009. https://www.nao.org.uk/reports/serv ices-for-people-with-rheumatoid-arthritis/ (15 September 2024, date last accessed).
- 10. Scott IC, Whittle R, Bailey J *et al.* Rheumatoid arthritis, psoriatic arthritis, and axial spondyloarthritis epidemiology in England from 2004 to 2020: an observational study using primary care electronic health record data. Lancet Reg Health Eur 2022;23:100519.

- 11. Costello R, McDonagh J, Hyrich KL, Humphreys JH. Incidence and prevalence of juvenile idiopathic arthritis in the United Kingdom, 2000–2018: results from the Clinical Practice Research Datalink. Rheumatol Oxf Engl 2022;61:2548–54.
- 12. Costello RE, Kearsley-Fleet L, McDonagh JE, Hyrich KL, Humphreys JH. Continuing specialist care into adulthood in young people with juvenile idiopathic arthritis: a retrospective cohort study using electronic health records in England. Rheumatology 2023;62:1936–43.
- 13. Ravelli A, Consolaro A, Horneff G *et al.* Treating juvenile idiopathic arthritis to target: recommendations of an international task force. Ann Rheum Dis 2018;77:819–28.
- 14. Vermeer M, Kuper HH, Moens HJB *et al.* Sustained beneficial effects of a protocolized treat-to-target strategy in very early rheumatoid arthritis: three-year results of the Dutch Rheumatoid Arthritis Monitoring remission induction cohort. Arthritis Care Res 2013;65:1219–26.
- Radner H, Smolen JS, Aletaha D. Remission in rheumatoid arthritis: benefit over low disease activity in patient-reported outcomes and costs. Arthritis Res Ther 2014;16:R56.
- 16. Jansen JP, Buckley F, Dejonckheere F, Ogale S. Comparative efficacy of biologics as monotherapy and in combination with methotrexate on patient reported outcomes (PROs) in rheumatoid arthritis patients with an inadequate response to conventional DMARDs—a systematic review and network meta-analysis. Health Qual Life Outcomes 2014;12:102.
- 17. Lopez-Olivo MA, Siddhanamatha HR, Shea B *et al*. Methotrexate for treating rheumatoid arthritis. Cochrane Database Syst Rev 2014;2014:CD000957.
- 18. Ibrahim F, Ma M, Scott DL, Scott IC. Defining the relationship between pain intensity and disease activity in patients with rheumatoid arthritis: a secondary analysis of six studies. Arthritis Res Ther 2022;24:218.
- McWilliams DF, Dawson O, Young A *et al*. Discrete trajectories of resolving and persistent pain in people with rheumatoid arthritis despite undergoing treatment for inflammation: results from three UK cohorts. J Pain 2019;20:716–27. Jun
- Scott IC, Ibrahim F, Lewis CM, Scott DL, Strand V. Impact of intensive treatment and remission on health-related quality of life in early and established rheumatoid arthritis. RMD Open 2016;2:e000270.
- 21. Carpenter L, Barnett R, Mahendran P *et al*. Secular changes in functional disability, pain, fatigue and mental well-being in early rheumatoid arthritis. A longitudinal meta-analysis. Semin Arthritis Rheum 2019;50:209–19.
- 22. Bromberg MH, Connelly M, Anthony KK, Gil KM, Schanberg LE. Self-reported pain and disease symptoms persist in juvenile idiopathic arthritis despite treatment advances: an electronic diary study. Arthritis Rheumatol 2014;66:462–9.
- Baerwald C, Stemmler E, Gnüchtel S *et al.* Predictors for severe persisting pain in rheumatoid arthritis are associated with pain origin and appraisal of pain. Ann Rheum Dis 2024;83:1381–8. https://ard.bmj.com/content/early/2024/05/30/ard-2023-225414 (28 June 2024, date last accessed).
- Rashid A, Cordingley L, Carrasco R *et al.* Patterns of pain over time among children with juvenile idiopathic arthritis. Arch Dis Child 2018;103:437–43.
- Courvoisier DS, Agoritsas T, Glauser J *et al.*; National Data Bank for Rheumatic Diseases. Pain as an important predictor of psychosocial health in patients with rheumatoid arthritis. Arthritis Care Res 2012;64:190–6.
- Wolfe F, Hawley DJ. The longterm outcomes of rheumatoid arthritis: work disability: a prospective 18 year study of 823 patients. J Rheumatol 1998;25:2108–17.
- Conaghan PG, Alten R, Deodhar A *et al.* Relationship of pain and fatigue with health-related quality of life and work in patients with psoriatic arthritis on TNFi: results of a multi-national real-world study. RMD Open 2020;6:e001240.
- 28. Hirano F, van der Heijde D, van Gaalen FA *et al*. Determinants of the patient global assessment of well-being in early axial

spondyloarthritis: 5-year longitudinal data from the DESIR cohort. Rheumatol Oxf Engl 2021;60:316–21.

- 29. Euesden J, Matcham F, Hotopf M *et al*. The relationship between mental health, disease severity, and genetic risk for depression in early rheumatoid arthritis. Psychosom Med 2017;79:638–45.
- Pollard LC, Choy EH, Gonzalez J, Khoshaba B, Scott DL. Fatigue in rheumatoid arthritis reflects pain, not disease activity. Rheumatology 2006;45:885–9.
- 31. Gossec L, de Wit M, Kiltz U *et al.*; EULAR PsAID Taskforce. A patient-derived and patient-reported outcome measure for assessing psoriatic arthritis: elaboration and preliminary validation of the Psoriatic Arthritis Impact of Disease (PsAID) questionnaire, a 13-country EULAR initiative. Ann Rheum Dis 2014;73:1012–9.
- 32. Heiberg T, Kvien TK. Preferences for improved health examined in 1,024 patients with rheumatoid arthritis: pain has highest priority. Arthritis Rheum 2002;47:391–7.
- 33. Turid H, Elisabeth L, Désirée van der H, Tore KK. Sleep problems are of higher priority for improvement for patients with ankylosing spondylitis than for patients with other inflammatory arthropathies. Ann Rheum Dis 2011;70:872–3.
- Scott IC, Machin A, Mallen CD, Hider SL. The extra-articular impacts of rheumatoid arthritis: moving towards holistic care. BMC Rheumatol 2018;2:32.
- Duffield SJ, Miller N, Zhao S, Goodson NJ. Concomitant fibromyalgia complicating chronic inflammatory arthritis: a systematic review and meta-analysis. Rheumatology 2018;57:1453–60.
- 36. Huang Y-T, Jenkins DA, Yimer BB *et al.* Trends for opioid prescribing and the impact of the COVID-19 pandemic in patients with rheumatic and musculoskeletal diseases between 2006 and 2021. Rheumatology 2024;63:1093–103.
- Hazlewood G, van der Heijde DM, Bombardier C. Paracetamol for the management of pain in inflammatory arthritis: a systematic literature review. J Rheumatol Suppl 2012;90:11–6.
- Paglia MDG, Silva MT, Lopes LC *et al.* Use of corticoids and nonsteroidal anti-inflammatories in the treatment of rheumatoid arthritis: systematic review and network meta-analysis. PLoS One 2021;16:e0248866.
- 39. Castellsague J, Riera-Guardia N, Calingaert B *et al.*; Safety of Non-Steroidal Anti-Inflammatory Drugs (SOS) Project. Individual NSAIDs and upper gastrointestinal complications: a systematic review and meta-analysis of observational studies (the SOS project). Drug Saf 2012;35:1127–46.
- Versus Arthritis. Unseen, Unequal and Unfair: Chronic Pain in England. https://www.versusarthritis.org/news/2021/june/unseen-unequal-and-un fair-chronic-pain-in-england/ (5 May 2024, date last accessed).
- Population Health Directorate of the Scottish Government. The Scottish Health Survey 2022: summary report. http://www.gov. scot/publications/scottish-health-survey-2022-summary-report/ (11 June 2024, date last accessed).
- 42. Taylor S, Annand F, Burkinshaw P et al. Dependence and withdrawal associated with some prescribed medicines: an evidence review. London: Public Health England 2019. https://assets. publishing.service.gov.uk/media/5fc658398fa8f5474c800149/PHE\_ PMR\_report\_Dec2020.pdf (18 October 2024, date last accessed).
- 43. Jani M, Birlie Yimer B, Sheppard T, Lunt M, Dixon WG. Time trends and prescribing patterns of opioid drugs in UK primary care patients with non-cancer pain: a retrospective cohort study. PLoS Med 2020;17:e1003270.
- ARMA. Reduce health inequalities in musculoskeletal health. https://arma.uk.net/msk-health-inequalities-equality/ (11 June 2024, date last accessed).
- NICE. Recommendations | Rheumatoid arthritis in adults: management | Guidance | NICE. 2018. https://www.nice.org.uk/guid ance/ng100/chapter/Recommendations (10 September 2024, date last accessed).
- 46. NICE. Recommendations | Spondyloarthritis in over 16s: diagnosis and management | Guidance | NICE. 2017. https://www.nice. org.uk/guidance/ng65/chapter/Recommendations (September 2024, date last accessed).

- Geenen R, Overman CL, Christensen R *et al.* EULAR recommendations for the health professional's approach to pain management in inflammatory arthritis and osteoarthritis. Ann Rheum Dis 2018; 77:797–807.
- NICE. Overview | Chronic pain (primary and secondary) in over 16s: assessment of all chronic pain and management of chronic primary pain | Guidance | NICE [Internet]. 2021. https://www.nice. org.uk/guidance/NG193 (5 May 2024, date last accessed).
- 49. Raja SN, Carr DB, Cohen M *et al.* The revised International Association for the Study of Pain definition of pain: concepts, challenges, and compromises. Pain 2020;161:1976–82.
- International Association for the Study of Pain (IASP). Terminology | International Association for the Study of Pain [Internet]. https://www. iasp-pain.org/resources/terminology/ (12 June 2024, date last accessed).
- 51. Chapman LS, Backhouse M, Bearne L et al.; British Society for Rheumatology Standards, Audit and Guidelines Working Group. Management of foot health in people with inflammatory arthritis: British Society for Rheumatology guideline scope. Rheumatology 2022;61:3907–11.
- 52. Ndosi M, Ferguson R, Backhouse MR *et al.* National variation in the composition of rheumatology multidisciplinary teams: a cross-sectional study. Rheumatol Int 2017;37:1453–9.

© The Author(s) 2024. Published by Oxford University Press on behalf of the British Society for Rheumatology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited. Rheumatology Advances in Practice, 2024, 8, 1–7 https://doi.org/10.1093/rap/rkae128

This promotional material has been created and funded by Novartis Pharmaceuticals UK Ltd. for UK healthcare professionals only. Prescribing information can be found on the next page. Adverse event statement found below.

Biologics may be **less** effective in patients who are overweight<sup>1,2</sup>

Eligible patients, weighing ≥90kg with PsA and concomitant moderate to severe PsO, may need an individualised treatment approach<sup>4,5</sup>

**>6 in 10** adults over the age of 18 years in England are estimated to be overweight or living with obesity<sup>3</sup>

Ü

Click here to visit our HCP portal and learn more

Cosentyx<sup>®</sup> (secukinumab) provides **flexible dosing** based on your eligible patients' needs<sup>\*4,5</sup>



Adapted from Cosentyx<sup>®</sup> (secukinumab) SmPC.<sup>4,5</sup>

\*For adult patients with PsA and concomitant moderate to severe PsO, the recommended dose of Cosentyx is 300 mg with initial dosing at Weeks 0, 1, 2, 3 and 4, followed by **monthly maintenance dosing**. Based on clinical response, a maintenance dose of 300 mg **Q2W** may provide additional benefit for patients with a body weight of **90 kg or higher**.<sup>4,5</sup>

Cosentyx is indicated for the treatment of moderate to severe plaque psoriasis in adults, children and adolescents from the age of 6 years who are candidates for systemic therapy; active psoriatic arthritis in adult patients (alone or in combination with methotrexate) when the response to previous disease-modifying anti-rheumatic drug therapy has been inadequate; active ankylosing spondylitis in adults who have responded inadequately to conventional therapy; active non-radiographic axial spondyloarthritis with objective signs of inflammation as indicated by elevated C-reactive protein and/or magnetic resonance imaging evidence in adults who have responded inadequately to non-steroidal anti-inflammatory drugs; active moderate to severe hidradenitis suppurativa (acne inversa) in adults with an inadequate response to conventional systemic therapy; active enthesitis-related arthritis in patients 6 years and older (alone or in combination with methotrexate) whose disease has responded inadequately to, or who cannot tolerate, conventional therapy; active juvenile psoriatic arthritis in patients 6 years and older (alone or in combination with methotrexate) whose disease has responded inadequately to, or who cannot tolerate, conventional therapy; 4.5

PsA, psoriatic arthritis; PsO, plaque psoriasis; Q2W, every 2 weeks.

**References: 1.** Warren RB, et al. *J Invest Dermatol* 2015;135:2632–2640; **2.** Warren RB, et al. *Br J Dermatol* 2019;180(5):1069–1076; **3.** Office for Health Improvement and Disparities. Obesity profile: short statistical commentary May 2024. Available at: https://www.gov.uk/government/statistics/update-to-the-obesity-profile-on-fingertips/ obesity-profile-short-statistical-commentary-may-2024 [Accessed August 2024]; **4.** Cosentyx<sup>®</sup> (secukinumab) GB Summary of Product Characteristics; **5.** Cosentyx<sup>®</sup> (secukinumab) NI Summary of Product Characteristics.

Adverse events should be reported. Reporting forms and information can be found at www.mhra.gov.uk/yellowcard. Adverse events should also be reported to Novartis online through the pharmacovigilance intake (PVI) tool at www.novartis.com/report, or alternatively email medinfo.uk@novartis.com or call 01276 698370

# **U**NOVARTIS

#### Cosentyx<sup>®</sup> (secukinumab) Great Britain Prescribing Information.

# Please refer to the Summary of Product Characteristics (SmPC) before prescribing.

Indications: Treatment of: moderate to severe plague psoriasis in adults children and adolescents from the age of 6 years who are candidates for systemic therapy; active psoriatic arthritis in adults (alone or in combination with methotrexate) who have responded inadequately to disease-modifying anti-rheumatic drug therapy: active ankylosing spondylitis in adults who have responded inadequately to conventional therapy: active non-radiographic axial spondyloarthritis (nr-axSpA) with objective signs of inflammation as indicated by elevated C-reactive protein (CRP) and/or magnetic resonance imaging (MRI) evidence in adults who have responded inadequately to non-steroidal anti-inflammatory drugs; active enthesitis-related arthritis and juvenile psoriatic arthritis in patients 6 years and older (alone or in combination with methotrexate) whose disease has responded inadequately to, or who cannot tolerate, conventional therapy; active moderate to severe hidradenitis suppurativa (acne inversa) in adults with an inadequate response to conventional systemic HS therapy. Presentations: Cosentyx 75 mg solution for injection in pre-filled syringe; Cosentyx 150 mg solution for injection in pre-filled syringe; Cosentyx 150 mg solution for injection in pre-filled pen; Cosentyx 300 mg solution for injection in pre-filled pen. Dosage & Administration: Administered by subcutaneous injection at weeks 0, 1, 2, 3 and 4, followed by monthly maintenance dosing. Consider discontinuation if no response after 16 weeks of treatment. Each 75 mg dose is given as one injection of 75 mg. Each 150 mg dose is given as one injection of 150 mg. Each 300 mg dose is given as two injections of 150 mg or one injection of 300 mg. If possible avoid areas of the skin showing psoriasis. Plaque Psoriasis: Adult recommended dose is 300 mg. Based on clinical response, a maintenance dose of 300 mg every 2 weeks may provide additional benefit for patients with a body weight of 90 kg or higher. Adolescents and children from the age of 6 years: if weight  $\geq$  50 kg. recommended dose is 150 mg (may be increased to 300 mg as some patients may derive additional benefit from the higher dose). If weight < 50 kg recommended dose is 75 mg. *Psoriatic Arthritis*: For patients with concomitant moderate to severe plaque psoriasis see adult plaque psoriasis recommendation. For patients who are anti-TNFa inadequate responders, the recommended dose is 300 mg, 150 mg in other patients. Can be increased to 300 mg based on clinical response. Ankylosing Spondylitis: Recommended dose 150 mg. Can be increased to 300 mg based on clinical response. nr-axSpA: Recommended dose 150 mg. Enthesitis-related arthritis and juvenile psoriatic arthritis: From the age of 6 years, if weight ≥ 50 kg, recommended dose is 150 mg. If weight < 50 kg, recommended dose is 75 mg. Hidradenitis suppurativa:

## Cosentyx® (secukinumab) Northern Ireland Prescribing Information.

## Please refer to the Summary of Product Characteristics (SmPC) before prescribing.

Indications: Treatment of: moderate to severe plaque psoriasis in adults. children and adolescents from the age of 6 years who are candidates for systemic therapy; active psoriatic arthritis in adults (alone or in combination with methotrexate) who have responded inadequately to disease-modifying anti-rheumatic drug therapy; active ankylosing spondylitis in adults who have responded inadequately to conventional therapy; active non-radiographic axial spondyloarthritis (nr-axSpA) with objective signs of inflammation as indicated by elevated C-reactive protein (CRP) and/or magnetic resonance imaging (MRI) evidence in adults who have responded inadequately to non-steroidal anti-inflammatory drugs; active enthesitis-related arthritis and juvenile psoriatic arthritis in patients 6 years and older (alone or in combination with methotrexate) whose disease has responded inadequately to, or who cannot tolerate, conventional therapy; active moderate to severe hidradenitis suppurativa (acne inversa) in adults with an inadequate response to conventional systemic HS therapy. Presentations: Cosentyx 150 mg solution for injection in pre-filled pen; Cosentyx 300 mg solution for injection in pre-filled pen. Dosage & Administration: Administered by subcutaneous injection at weeks 0, 2, 3 and 4, followed by monthly maintenance dosing. Consider discontinuation if no response after 16 weeks of treatment. Each 150 mg dose is given as one injection of 150 mg. Each 300 mg dose is given as two injections of 150 mg or one injection of 300 mg. If possible avoid areas of the skin showing psoriasis. Plaque Psoriasis: Adult recommended dose is 300 mg monthly. Based on clinical response, a maintenance dose of 300 mg every 2 weeks may provide additional benefit for patients with a body weight of 90 kg or higher. Adolescents and children from the age of 6 years; if weight  $\geq$  50 kg. recommended dose is 150 mg (may be increased to 300 mg as some patients may derive additional benefit from the higher dose). If weight < 50 kg, recommended dose is 75 mg. However, 150mg solution for injection in pre-filled pen is not indicated for administration of this dose and no suitable alternative formulation is available. Psoriatic Arthritis: For patients with concomitant moderate to severe plaque psoriasis see adult plaque psoriasis recommendation. For patients who are anti-TNFa inadequate responders, the recommended dose is 300 mg, 150 mg in other patients. Can be increased to 300 mg based on clinical response. Ankylosing Spondylitis: Recommended dose 150 mg. Can be increased to 300 mg based on clinical response. nr-axSpA: Recommended dose 150 mg. Enthesitis-related arthritis and juvenile psoriatic arthritis: From the age of 6 years, if weight ≥ 50 kg, recommended dose is 150 mg. If weight < 50 kg, recommended dose

Recommended dose is 300 mg monthly. Based on clinical response, the maintenance dose can be increased to 300 mg every 2 weeks. Contraindications: Hypersensitivity to the active substance or excipients, Clinically important, active infection, Warnings & Precautions: Infections: Potential to increase risk of infections; serious infections have been observed. Caution in patients with chronic infection or history of recurrent infection. Advise patients to seek medical advice if signs/symptoms of infection occur. Monitor patients with serious infection closely and do not administer Cosentyx until the infection resolves. Non-serious mucocutaneous candida infections were more frequently reported for secukinumab in the psoriasis clinical studies. Should not be given to patients with active tuberculosis (TB) Consider anti-tuberculosis therapy before starting Cosentyx in patients with latent TB. Inflammatory bowel disease (including Crohn's disease and ulcerative colitis): New cases or exacerbations of inflammatory bowel disease have been reported with secukinumab. Secukinumab. is not recommended in patients with inflammatory bowel disease. If a natient develops signs and symptoms of inflammatory bowel disease or experiences an exacerbation of pre-existing inflammatory bowel disease secukinumab should be discontinued and appropriate medical management should be initiated. Hypersensitivity reactions: Rare cases of anaphylactic reactions have been observed. If an anaphylactic or serious allergic reactions occur, discontinue immediately and initiate appropriate therapy. Vaccinations: Do not give live vaccines concurrently with Cosentyx: inactivated or non-live vaccinations may be given. Paediatric patients should receive all age appropriate immunisations before treatment with Cosentyx. Latex-Sensitive Individuals: The removable needle cap of the 75mg and 150 mg pre-filled syringe and 150mg pre-filled pen contains a derivative of natural rubber latex. Concomitant immunosuppressive therapy: Combination with immunosuppressants, including biologics, or phototherapy has not been evaluated in psoriasis studies. Cosentyx was given concomitantly with methotrexate, sulfasalazine and/or corticosteroids in arthritis studies. Caution when considering concomitant use of other immunosuppressants. Interactions: Live vaccines should not be given concurrently with secukinumab. No interaction between Cosentyx and midazolam (CYP3A4 substrate) seen in adult psoriasis study. No interaction between Cosentyx and methotrexate and/or corticosteroids seen in arthritis studies. Fertility, pregnancy and lactation: Women of childbearing potential: Use an effective method of contraception during and for at least 20 weeks after treatment. Pregnancy: Preferably avoid use of Cosentyx in pregnancy. Breast feeding: It is not known if secukinumab is excreted in human breast milk. A clinical decision should be made on continuation of breast feeding during Cosentyx treatment (and up to 20 weeks after discontinuation) based on benefit of breast feeding to the child and benefit of Cosentyx therapy to the

is 75 mg. However, 150mg solution for injection in pre-filled pen is not indicated for administration of this dose and no suitable alternative formulation is available. Hidradenitis suppurativa: Recommended dose is 300 mg monthly. Based on clinical response, the maintenance dose can be increased to 300 mg every 2 weeks. Contraindications: Hypersensitivity to the active substance or excinients Clinically important, active infection. Warnings & Precautions: Infections: Potential to increase risk of infections: serious infections have been observed. Caution in patients with chronic infection or history of recurrent infection. Advise patients to seek medical advice if signs/ symptoms of infection occur. Monitor patients with serious infection closely and do not administer Cosentyx until the infection resolves. Non-serious mucocutaneous candida infections were more frequently reported for secukinumab than placebo in the psoriasis clinical studies Should not be given to patients with active tuberculosis (TB). Consider anti-tuberculosis therapy before starting Cosentyx in patients with latent TB. Inflammatory bowel disease (including Crohn's disease and ulcerative colitis): New cases or exacerbations of inflammatory bowel disease have been reported with secukinumab. Secukinumab is not recommended in patients with inflammatory bowel disease. If a patient develops signs and symptoms of inflammatory bowel disease or experiences an exacerbation of pre-existing inflammatory bowel disease, secukinumab should be discontinued and appropriate medical management should be initiated. Hypersensitivity reactions: Rare cases of anaphylactic reactions have been observed. If an anaphylactic or serious allergic reactions occur, discontinue immediately and initiate appropriate therapy. Vaccinations: Do not give live vaccines concurrently with Cosentyx; inactivated or non-live vaccinations may be given. Paediatric patients should receive all age appropriate immunisations before treatment with Cosentyx. Latex-Sensitive Individuals: The removable needle cap of the 150mg pre-filled pen contains a derivative of natural rubber latex. Concomitant immunosuppressive therapy: Combination with immunosuppressants, including biologics, or phototherapy has not been evaluated in psoriasis studies. Cosentyx was given concomitantly with methotrexate, sulfasalazine and/or corticosteroids in arthritis studies. Caution when considering concomitant use of other immunosuppressants. Interactions: Live vaccines should not be given concurrently with secukinumab. No interaction between Cosentyx and midazolam (CYP3A4 substrate) seen in adult psoriasis study. No interaction between Cosentyx and methotrexate and/or corticosteroids seen in arthritis studies. Fertility, pregnancy and lactation: Women of childbearing potential: Use an effective method of contraception during and for at least 20 weeks after treatment. Pregnancy: Preferably avoid use of Cosentyx in pregnancy. Breast feeding: It is not known if secukinumab is excreted in human breast milk. A clinical decision should be made on

woman. Fertility: Effect on human fertility not evaluated. Adverse **Reactions:** Very Common ( $\geq 1/10$ ): Upper respiratory tract infection. *Common* ( $\geq 1/100$  to <1/10): Oral herpes, headache, rhinorrhoea, diarrhoea, nausea, fatique. Uncommon (≥1/1,000 to <1/100): Oral candidiasis, lower respiratory tract infections, neutropenia, inflammatory bowel disease. Rare (≥1/10,000 to <1/1,000): anaphylactic reactions, exfoliative dermatitis (psoriasis patients), hypersensitivity vasculitis. Not known: Mucosal and cutaneous candidiasis (including oesophageal candidiasis). Infections: Most infections were non-serious and mild to moderate upper respiratory tract infections, e.g. nasopharyngitis, and did not necessitate treatment discontinuation. There was an increase in mucosal and cutaneous (including oesophageal) candidiasis, but cases were mild or moderate in severity, non-serious, responsive to standard treatment and did not necessitate treatment discontinuation. Serious infections occurred in a small proportion of patients (0.015 serious infections reported per patient year of follow up). Neutropenia: Neutropenia was more frequent with secukinumab than placebo, but most cases were mild, transient and reversible. Rare cases of neutropenia CTCAF Grade 4 were reported. Hypersensitivity reactions: Urticaria and rare cases of anaphylactic reactions were seen. Immunogenicity: Less than 1% of patients treated with Cosentyx developed antibodies to secukinumab up to 52 weeks of treatment. Other Adverse Effects: The list of adverse events is not exhaustive, please consult the SmPC for a detailed listing of all adverse events before prescribing. Legal Category: POM. MĂ Number & List Price: PLGB 00101/1205 - 75 mg pre-filled syringe - £304.70; PLGB 00101/1029 - 150 mg pre-filled pen x2 £1,218.78; PLGB 00101/1030 - 150 mg pre-filled syringe x2 £1,218.78; PLGB 00101/1198 - 300 mg pre-filled pen x 1 £1218.78. PI Last Revised: June 2023. Full prescribing information, (SmPC) is available from: Novartis Pharmaceuticals UK Limited, 2nd Floor, The WestWorks Building, White City Place, 195 Wood Lane, London, W12 7FQ. Telephone: (01276) 692255.

UK | 290802 | June 2023

#### Adverse Event Reporting:

Adverse events should be reported. Reporting forms and information can be found at <u>www.mhra.gov.uk/yellowcard</u>. Adverse events should also be reported to Novartis via uk.patientsafety@novartis.com or online through the pharmacovigilance intake (PVI) tool at <u>www.novartis.com/report</u>. If you have a question about the product, please contact Medical Information on 01276 698370 or by email at medinfo.uk@novartis.com

continuation of breast feeding during Cosentyx treatment (and up to 20 weeks after discontinuation) based on benefit of breast feeding to the child and benefit of Cosentyx therapy to the woman. Fertility: Effect on human fertility not evaluated. Adverse Reactions: Very Common (≥1/10): Upper respiratory tract infection. Common (≥1/100 to <1/10): Oral herpes, headache, rhinorrhoea, diarrhoea, nausea, fatigue. Uncommon (≥1/1,000 to <1/100): Oral candidiasis, lower respiratory tract infections, neutropenia, inflammatory bowel disease. Rare (≥1/10,000 to <1/1,000): anaphylactic reactions, exfoliative dermatitis (psoriasis patients), hypersensitivity vasculitis. Not known: Mucosal and cutaneous candidiasis (including oesophageal candidiasis). Infections: Most infections were non-serious and mild to moderate upper respiratory tract infections, e.g. nasopharyngitis, and did not necessitate treatment discontinuation. There was an increase in mucosal and cutaneous (including oesophageal) candidiasis, but cases were mild or moderate in severity, non-serious, responsive to standard treatment and did not necessitate treatment discontinuation. Serious infections occurred in a small proportion of patients (0.015 serious infections reported per patient year of follow up). Neutropenia: Neutropenia was more frequent with secukinumab than placebo, but most cases were mild, transient and reversible. Rare cases of neutropenia CTCAE Grade 4 were reported. <u>Hypersensitivity reactions</u>: Urticaria and rare cases of anaphylactic reactions were seen. Immunogenicity: Less than 1% of patients treated with Cosentyx developed antibodies to secukinumab up to 52 weeks of treatment. Other Adverse Effects: The list of adverse events is not exhaustive, please consult the SmPC for a detailed listing of all adverse events before prescribing. Legal Category: POM. MA Number & List Price: FU/1/14/980/005 150 mg pre-filled pen x2 £1.218.78 EU/1/14/980/010 - 300 mg pre-filled pen x 1 £1218.78. PI Last Revised: May 2023. Full prescribing information, (SmPC) is available from: Novartis Pharmaceuticals UK Limited, 2nd Floor, The WestWorks Building, White City Place, 195 Wood Lane, London, W12 7FQ. Telephone: (01276) 692255.

## UK | 284832 | May 2023

#### Adverse Event Reporting:

Adverse events should be reported. Reporting forms and information can be found at <u>www.mhra.gov.uk/yellowcard</u>. Adverse events should also be reported to Novartis via uk.patientsafety@novartis.com or online through the pharmacovigilance intake (PVI) tool at www.novartis.com/report If you have a question about the product, please contact Medical Information on 01276 698370 or by email at medinfo.uk@novartis.com